Author's response to reviews

Title: Is abdominal wall contraction important for normal voiding in the female rat?

Authors:

Phillip P. Smith (pps smith@bcm.tmc.edu)
Christopher P. Smith (cps@bcm.tmc.edu)
Timothy B. Boone (tboone@bcm.tmc.edu)
George T. Somogyi (gsomogyi@bcm.tmc.edu)

Version: Date: 8 26 February 2007

Author's response to reviews:

To the Editor,

Thank you for accepting this manuscript for publication. The requested formatting revisions have been made and the revised manuscript is now submitted.

Phillip P. Smith MD
Scott Department of Urology
Baylor College of Medicine
Houston TX